\documentclass[11pt,twoside]{article}
\usepackage{amsmath, amsthm, amscd, amsfonts, amssymb, graphicx, color}
\usepackage[bookmarksnumbered, colorlinks]{hyperref} \usepackage{float}
\usepackage{lipsum}
\usepackage{afterpage}
\usepackage[labelfont=bf]{caption}
\usepackage[nottoc,notlof,notlot]{tocbibind} 
%\renewcommand\bibname{References}
\def\bibname{\Large \bf  References}
\usepackage{lipsum}
\usepackage{fancyhdr}
\pagestyle{fancy}
\fancyhf{}
\renewcommand{\headrulewidth}{0pt}
\fancyhead[LE,RO]{\thepage}
\thispagestyle{empty}
%\afterpage{\lhead{new value}}

\fancyhead[CE]{} 
\fancyhead[CO]{Some new results on regular modules}



%\topmargin=-1.6cm
\textheight 17.5cm%
\textwidth  12cm %
\topmargin   8mm  %
\oddsidemargin   20mm   %
\evensidemargin   20mm   %
\footskip=24pt     %

\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{corollary}[theorem]{Corollary}
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{xca}[theorem]{Exercise}
%\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\renewenvironment{proof}{{\bfseries \noindent Proof.}}{~~~~$\square$}
\makeatletter
\def\th@newremark{\th@remark\thm@headfont{\bfseries}}
\makeatletter



  
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% If you want to insert other packages. Insert them here
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%\long\def\symbolfootnote[#1]#2{\begingroup%
%\def\thefootnote{\fnsymbol{footnote}}\footnote[#1]{#2}\endgroup}



 \def \thesection{\arabic{section}}
 

\begin{document}
\begin{thebibliography}{99} % Enter references in alphabetical order and according to the following format.
\bibitem{chenn} A. M. Abduldaim and S. Chen,   GF-regular modules,  {\it J. Appl. Math.}, (2013) 7 pages.


\bibitem{09Amiri}
N. Amiri,  M. Ershad, On regular module, {\it International Journal of Pure and Applied Mathematics}, 19 (2005) 43-47.
	
\bibitem{02amiri}
N. Amiri, Some results on regular modules, {\it Journal of mathematical extension}, 2 (2007-2008)  71-79.

\bibitem{08Anderson}
D. D. Anderson, C. Sangmin, J. R. Juett,
Module-theoretic generalization of commutative von Neumann regular rings, {\it Comm. Algebra}, 47 (2019) 4713-4728.
	


	
	
\bibitem{11Courter[6]} R. C. Courter,  Rings all of whose factor rings are semi-prime, {\it Canad. Math. Bull.}, 12  (1969) 417-426. 


\bibitem{12Fisher[8]}J. W. Fisher,   Von Neumann regular rings versus V-rings, In Rings Theory, {\it Proc. Conf. Univ. Oklahoma, Norman Oklahoma},  New York: Dekker, (1974) 101-119.


	\bibitem{03Goodearl}
K. R. Goodearl,  Von Neumann regular rings. Vol. 4. London: Pitman, 1979.



\bibitem{14Hansen[9]} F. Hansen,   On one-side prime ideals, 
 {\it Pacific J. Math}, 58 (1975) 79-85. 
	
	
	\bibitem{06hassanzade}
	D. Hassanzadeh-Lelekaami,  H. Roshan-Shekalgourabi,On regular modules over commutative rings, {\it  Bulletin of the Malaysian Mathematical Sciences Society}, 42 (2019) 569-583.
	
	
	
\bibitem{07Jayaram}
C. Jayaram, C., Tekir, {\"U}nsal
 von Neumann regular modules
 {\it Comm. Algebra}, 46 (2018) 2205-2217.




\bibitem{10koh[11]} K. Koh,  On one-sided ideals of a prime type,  {\it Proc. Amer. Math. Soc.}, 28   (1971) 321-329. 



	
	\bibitem{01McCasland}
	R. L. McCasland, Patrick F. Smith, {On isolated submodules,} {\it Comm. Algebra}, 34(8) (2006)  2977-2988.
	


	\bibitem{05Ware}
R. Ware, Endomorphism rings of projective modules,
 {\it Trans. Amer.  Math. Soc.}, 155 (1971) 233-256.



	\bibitem{04Zelmanowitz}
J. Zelmanowitz, Regular modules,
 {\it Trans. Amer.  Math. Soc.}, 163 (1972) 341-355.






\end{thebibliography}

%\baselineskip 9mm
%\setcounter{page}{}
\thispagestyle{plain}
{\noindent Journal of Mathematical Extension \\
Vol. XX, No. XX, (2014), pp-pp (Will be inserted by layout editor)}\\
ISSN: 1735-8299\\
URL: http://www.ijmex.com\\
‎\vspace*{9mm}
‎
\begin{center}

{\Large \bf 
Some new results on regular modules\\}  
{} 


\let\thefootnote\relax\footnote{\scriptsize Received: XXXX; Accepted: XXXX (Will be inserted by editor)}

{\bf Author}\vspace*{-2mm}\\
\vspace{2mm} {\small  } \vspace{2mm}

{\bf  Author\let\thefootnote\relax\footnote{$^*$Corresponding Author}}\vspace*{-2mm}\\
\vspace{2mm} {\small   } \vspace{2mm}

\end{center}

\vspace{4mm}


{\footnotesize
\begin{quotation}
{\noindent \bf Abstract.} 
		  		  The aim of this paper is to study unitary regular modules on commutative rings with identity. Regularity accompanied by cocyclic property results in some prime-related conclusions on both modules and rings. Further to this, regularity addresses also radical property of submodules and they are related closely. This property not only affects the modules on ring $R$ but also restricts R to totally idempotent one.

\end{quotation}
\begin{quotation}
\noindent{\bf AMS Subject Classification:}  	13C05

\noindent{\bf Keywords and Phrases:} Regular module; Pure submodule; Prime submodule; Primary submodule; Isolated submodule.
\end{quotation}}

\section{Introduction}
\label{intro}

Regularity concept started with a paper \cite{03Goodearl} of Von Neumann for rings with the regular property for elements and then was extended to modules in \cite{05Ware,04Zelmanowitz}. This concept has been studied widely in  papers and is in interest \cite{09Amiri,08Anderson,06hassanzade,07Jayaram}.

In \cite{02amiri}, it was proved that every projective module on Von Neumann regular ring is a regular module.
Moreover, it was proved a projective module $M$ is regular if and only if every cyclic submodule of $M$ is a direct summand of $M$. Also, they considered both regularity and projectivity properties and concluded $Z(M)=J(M)=0$, where $J(M)$ and $Z(M)$ are the Jacobson and singular submodules of $M$ \cite{02amiri}. Moreover, they proved $M$ is regular and indecomposable if and only if $M$ is projective and hollow.


In this paper, we would like to investigate regular modules with respect to prime, primary, idempotent and radical properties which lead us to interesting results as follows: 

Regular submodule not only has effects on its submodules but also it affects the ring; in regular $R$-module $M$, $R$ is simple if and only if every proper ideal is prime if and only if every submodule of $R$-module $M$ is prime.  In every regular $R$-module, $R$ is fully idempotent. We know that zero submodule can characterize many things, so knowing that it is prime, is important to us and in the realm of regularity, we get if $M$ is regular cocyclic $R$-module then  $0$ is  a prime submodule of $M$. Thinking of prime and primary properties is important because of their oneway relation but here we can give an equivalency between them; if $M$ is torsion-free and regular, then every proper submodule $N$ of $M$ is prime if and only if $N$ is a primary submodule of $M$.


Radical submodules are other ones to study under regularity. Here, we only give one result on them; every submodule of a regular $R$-module $M$ is a radical submodule of $M$.

We may like to go deeply and study submodules of submodules. In this investigation we get that for all proper submodule contained in $N$ there exists a prime  submodule of $N$ containing it.


 Moreover, by assuming that $R$ is regular we will get different results;  on every regular ring $R$, every submodule $N \neq M$ is a primary  submodule if and only if it is a  prime submodule of $M$.


Here are some basic definitions and results required for the rest of paper.


%\section{title}
\begin{definition}\cite[Definition 2.1]{02amiri}  \label{def1}
	Let $M$ be an $R$-module. An element $x\in M$ is called regular if there exists $\varphi\in {\rm Hom}_R(M, R)$ such that $\varphi(x)x=x$. The $R$-module $M$ is called regular if every element $x\in M$ is regular.
\end{definition}
\begin{definition} \label{def2}
	An $R$-submodule $N$ of $M$ is called pure if $N \cap IM=IN$ for every ideal $I$ of $R$.
\end{definition}
\begin{lemma}\cite[Lemma 1]{09Amiri}  \label{lemma1}
	If $M$ is a regular $R$-module, then every submodule of $M$ is pure.
\end{lemma}




\begin{lemma} \cite{chenn} \label{lemma3}
	A ring $R$ is Von Neumann regular if  and only if   $R$ is regular as an $R$-module.
\end{lemma}
\begin{definition} \cite{01McCasland} \label{def3}
	A submodule $ L $ of an $ R $-module $ M $ is called isolated if $ rad_ML \neq rad_MN $ for every proper submodule $ N $ of $ L $.
\end{definition} 
\begin{theorem}\label{lale11}\cite[Theorem 2.6]{01McCasland}
	The following statement are equivalent.
	\begin{itemize}
		\item Every submodule is an isolated submodule of $M$.
		\item For any given submodule $N$ of $M$ we have $N \cap IM=IN$ where $I$ is an arbitrary ideal of $R$.
	\end{itemize}
\end{theorem}




\section{Main results}
This section is devoted to proving the results  stated in the previous section. First of all, we bring a result on the isolated submodules.
  \begin{proposition} \cite{01McCasland}\label{pro1.1}
  	Suppose $ R $ is an arbitrary ring. Then for any proper submodule  $N$, we define it to be isolated  if and only if  every submodule $H$ of $N$ is contained in a prime submodule $ L $ of $ M $ such that  $ H \subseteq L  $ but $ N \nsubseteq L $. 
  \end{proposition}       
Here, we start with some results making the way to our results. Every isolated submodule is pure and the converse is also true, \cite[Theorem 1.9]{01McCasland}. So, every submodule of regular module is isolated. It is worth mentioning that   submodules of $R$-module $R$ which are prime, are also prime as ideals of $R$\cite{01McCasland}.  

\begin{theorem} \label{corollary1.3}
Let $M$ be a regular $R$-module. Then every proper submodule of $N$ is contained in a prime submodule of $N$.
\end{theorem}
\begin{proof} 
		Let $N$ be a submodule of a regular $R$-module $M$. Then from Lemma \ref{lemma1} $N$ is pure and by Theorem \ref{lale11} $N$ is an isolated submodule of $M$.  Proposition \ref{pro1.1} implies that for every submodule $H$ contained in $N$, there exists a prime submodule $ L $ of $ M $ in which $L$ involves $H$   but $ N$ is out of $L $ and so by \cite[Lemma 1.2]{01McCasland}  $L \cap N$ is  prime.
\end{proof}


The converse of above theorem dose not hold. Suppose I is a proper ideal of $R$ which is not direct summand of $R$. We know that every proper ideal $J$ of $I$ is containd in a maximal ideal and so in a prime ideal of $R$. However, it dose not imply that $R$ is a regular $R$-module. In \cite{01McCasland}, the authors brought an example of an ideal contained in a prime one but is not isolated and so is not pure claiming non-regularity of $R$.
\begin{proposition} \label{pro1.5}
	In regular $R$-module $M$,  every proper submodule $H$ of submodule $N$ is contained in  a prime submodule $K \subseteq N$,  in which $K$ has the property that
	can be lifted to $M$.
\end{proposition}
\begin{proof}
	We know that every submodule of regular module $M$ is pure, and so is isolated, according to Theorem \ref{lale11}.
	Let $H$ be the stated submodule in the proposition. By using  \cite[Proposition 1.6]{01McCasland}, the existence of such prime submodule $K$ of $N$ is trivial  such that $H\subseteq  K$ and satisfies the stated property.
\end{proof}

Now, one can try to understand under what conditions, a prime submodule $K$ can be lifted to $M$. Suppose $L$ is a prime submodule of an $R$-module $M$. Set $P=(L:M)$ which is clearly a prime ideal of $R$. In this case, $L$ is called $P$-prime submodule $M$. McCasland and Smith \cite{01McCasland} (Lying Over Theorem) proved that a $P$-prime submodule $K$ of $M$ can be lifted to $M$ if and only if $N \cap PM \subseteq  K$.

Now, if we consider Lying Over with regularity property plus the fact that in regular modules we have
$N \cap PM=PN$
we get the following corollary.
\begin{corollary}
Let $M$ be a regular $R$-module and $N$ be a $R-$submodule of $M$. Let $K$ be a $P-$prime submodule of $N$. Then $PN \subseteq K$  if and only if $K$ can be lifted to $M$.
\end{corollary}

\begin{corollary} \label{corollary1.7}
	For every  submodule $N$  of $M$, we have $IN=N$ if and only if $N \subseteq  IM$, where $I$ is an arbitrary ideal  of $R$.
\end{corollary}
\begin{proof}
	Suppose $N \subseteq  IM$. Then $N=N\cap IM$. On the other hand, $N$ is pure and we have $IN=N\cap IM$, where  $N=IN$. Conversely, let  $N=IN$. Then $N\cap IM= IN =N$ and we get $N \subseteq  IM$.
\end{proof}













\begin{theorem}\label{theorem1.8}
	Let $R$ be a regular ring and $M$ be an $R$-module. Then every proper submodule $N$ of $M$ is primary if and only if  is prime.
\end{theorem}
\begin{proof}
	If we suppose $N$ is a prime submodule, then it is a primary submodule. Conversely, suppose $N$ is a primary submodule of $M$. So $(N:M)$ is a primary ideal of $R$ and since $R$ is regular, every  primary ideal of $R$ is a prime ideal. Set $P=(N:M)$ and suppose that $rm \in N$ such that $m \notin N$. So $r^kM \subseteq  N$ which yields $r^k \in P$.
	From fact that $P$ is a prime ideal of $R$, then we get $r\in P$, which results  in $rM \subseteq  N$. So, we get $N$ is a prime submodule of $M$.
\end{proof}
\begin{theorem}\label{theorem1.9}
	Suppose $M$  is torsion-free and regular. Then every proper submodule $N$ of $M$ is prime if and only if $N$ is a primary submodule of $M$.
\end{theorem}
\begin{proof}
	The proof of the necessary condition is clear. For conversely, suppose $N$ is primary and $rx \in N$ for some $r\in R$, and $x\in M \setminus N$.
	It is enough to prove that $rM \subseteq  N$. Since $N$ is primary, then $r^kM\subseteq  N$ which accompanied by regularity of $M$ yields
	 $\left\langle r \right\rangle M \cap N=\left\langle r \right\rangle N$. Suppose $y \in M$. Then $r^k y=srn$ for some $s\in R, n \in N$, that gives us $r(r^{k-1}y-sn)=0$, then $r^{k-1}y=sn \in N$, hence $r^{k-1} M \subseteq  N$, so $r^{k-1}y= srn$, then $r^{k-2}y=sn \in N$ and by above argument $ry\in N$,
	 then $rM \subseteq  N$ so $N$ is a prime submodule of $M$.
\end{proof}

It is worthy of investigation relation between isolated submodule $N$ of $M$ and radical submodules of $M$ which are submodules of $N$, we know if it is a radical submodule of $M$ then $rad_M H=H\neq rad_M N$. It follows that $M$ is isolated if  all proper submodules of $N$ are  radical ones, also.
.


 
 \begin{corollary}\label{proposition2.1}
 Let $R$ be a regular ring and $M$ be an $R$-module.
 	The followings are equivalent.
 	\begin{itemize}
 		\item[(1)] $R$ is simple.
 		\item[(2)] Every  proper submodule $N$ of every non-zero $R$-module $M$ is prime.
 		\item[(3)] Every proper submodule $N$  of every non-zero $R$-module $M$ is primary.
 	\end{itemize}
 \end{corollary}
 \begin{proof}
 $(1) \Leftrightarrow (2)$ According to \cite[Theorem 4.2]{10koh[11]}.
 
 \noindent $(2) \Leftrightarrow (3)$ Theorem \ref{theorem1.8} gives this equivalency.
 \end{proof}
 
  \begin{lemma} \label{lemma2.5}
 Let $M$ be a regular  $R$-module which contains an essential simple submodule, i.e., $M$ is cocuclic module. Then $0$ is a prime submodule of $M$.
\end{lemma}
\begin{proof}
Let $IN=0$ and $N \neq 0$. We prove that $IM=0$ resulting   $0$ is a prime submodule of $M$. Let $L \neq 0$ be an essential simple submodule of $M$. Put $P=\{r \in R | rL=0     \}$, since $M$ is regular then $L \cap PM= PL=0$. So $PM=0$. On the other hand, $L \subseteq  N$ and $IN=0$. So, $IL=0$ which gives us $I \subseteq  P$ hence $IM \subseteq PM=0$  and so $IM=0$ and  we get the result.
\end{proof}


Let $R$ be a regular $R$-module. Then $I \cap J=I \cap RJ=IJ$. Put $I=J$. Then we have $I=I \cap I = II= I^2$. Hence, $R$ is fully idempotent(for definition of fully idempotent see \cite{11Courter[6], 12Fisher[8]}). So, in regular $R$-module $R$, $R$ is fully idempotent. In \cite[Propositions 2.2]{01McCasland}, the author proved that the ring $R$ is fully idempotent if and only if every submodule $N$ is pure submodule of $R$-module $M$. 

Hansen \cite[Lemma 1]{14Hansen[9]} proved that     the fully idempotent  ring $R$    is equivalent with the property that every proper ideals are radical submodules of $R$- module $R$. These explanations lead us to the following corollary.


\begin{corollary}
Let $R$ be a regular $R$-module. Then every proper submodule of $R$ is a radical submodule.
\end{corollary}

McCasland and Smith \cite[Propositions 2.6]{01McCasland} proved that $N$ is radical submodule of $M$ if and only if $N$ is a pure submodule of $M$. Since every submodule of a regular module is pure , so in every regular module, every submodule is both radical and pure.  As we see, the following result can be achieved.

\begin{corollary}
In every regular $R$-module $M$, all proper submodules of $M$ are radical and furthermore, $R$ is a fully idempotent ring.
\end{corollary}






\begin{center}

\end{center}



{\small

\noindent{\bf }

\noindent 

%\noindent Assistant Professor of Mathematics

\noindent 

\noindent 

\noindent E-mail: }\\

{\small
\noindent{\bf   }

\noindent  

\noindent 

\noindent  


\noindent 

\noindent E-mail: }\\



\end{document}